Author's response to reviews

Title: An analysis of policy levers used to implement mental health reform in Australia 1992-2012

Authors:

   Francesca C Grace (frankiechew@hotmail.com)
   Carla S Meurk (c.meurk@uq.edu.au)
   Brian Head (brian.head@uq.edu.au)
   Wayne D Hall (w.hall@uq.edu.au)
   Georgia Carstensen (georgia.carstensen@gmail.com)
   Meredith Harris (Meredith_harris@gcmhr.uq.edu.au)
   Harvey A Whiteford (h.whiteford@sph.uq.edu.au)

Version: 2 Date: 12 May 2015

Author's response to reviews:

1. Addition of title page
2. Addition of acknowledgements section
3. Addition of line numbers in main section of manuscript
4. NB: Ethics statement not applicable for this manuscript